

Format for ANSWERING REVIEWERS



February 15, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 9702-review.doc).

Title: Traumatic cataracts in children and its visual outcome

Author: Mehul A Shah, Shreya M Shah, Aarti H Chaudhry, Sandip Pannu

Name of Journal: *World Journal of Ophthalmology*

ESPS Manuscript NO: 9702

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

1. When we compared achieved visual outcome compared with ocular trauma score predicted vision.-corrected
2. Traumatic cataracts in children may have better outcome and ocular trauma score useful predictive method for the same. -corrected
3. At our centre in this study visual outcomes in eyes operated for cataracts which suffered trauma were analyzed. -corrected
4. . Bio statician certified data analyses report.
5. The achieved visual acuity after six weeks of surgery was $> 6/60$ in 450(82.7%) and $\geq 6/12$ 215 (39.4%) eyes in open globe group and $> 20/200$ in 127 (81.8%) and $\geq 6/1236$ (28.4%) eyes in Closed globe group (P = 0.143), and the difference between the groups was not significant in children. And visual acuity after six weeks of surgery was $> 6/60$ in 450(82.7%) and $\geq 6/12$ 215 (39.4%) eyes in open globe group and $> 20/200$ in 127 (81.8%) and $\geq 6/1236$ (28.4%) eyes in Closed globe group (P = 0.143), and the difference between the groups was not significant in children. and many other sentences Please give only metric (6/6) or foot system (20/20) in visual outcomes-corrected

Thank you again for publishing our manuscript in the *World Journal of Ophthalmology*.

Sincerely yours,

Mehul Shah
Drashti Netralaya
Chakalia Road

Dahod-389151

Gujarat

India

E-mail: omtrust@rediffmail.com